UC Davis UC Davis Previously Published Works

Title

Paradoxical myeloid-derived suppressor cell reduction in the bone marrow of SIV chronically infected macaques

Permalink https://escholarship.org/uc/item/0cn678g2

Journal PLOS Pathogens, 13(5)

ISSN 1553-7366

Authors

Sui, Yongjun Frey, Blake Wang, Yichuan <u>et al.</u>

Publication Date 2017

DOI

10.1371/journal.ppat.1006395

Peer reviewed



Citation: Sui Y, Frey B, Wang Y, Billeskov R, Kulkarni S, McKinnon K, et al. (2017) Paradoxical myeloid-derived suppressor cell reduction in the bone marrow of SIV chronically infected macaques. PLoS Pathog 13(5): e1006395. https:// doi.org/10.1371/journal.ppat.1006395

Editor: Guido Silvestri, Emory University, UNITED STATES

Received: February 8, 2017

Accepted: May 3, 2017

Published: May 12, 2017

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CCO public domain dedication.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This work is supported by NCI intramural funding ZO1 SC 004020. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Paradoxical myeloid-derived suppressor cell reduction in the bone marrow of SIV chronically infected macaques

Yongjun Sui¹*, Blake Frey¹, Yichuan Wang¹, Rolf Billeskov¹, Shweta Kulkarni¹, Katherine McKinnon¹, Tracy Rourke², Linda Fritts², Christopher J. Miller², Jay A. Berzofsky¹*

1 Vaccine Branch, National Cancer Institute, National Institutes of Health, Bethesda, MD, United States of America, 2 Center for Comparative Medicine, University of California Davis, Davis, CA, United States of America

* suiy@mail.nih.gov (YS); berzofsj@mail.nih.gov (JAB)

Abstract

Myeloid derived suppressor cells (MDSCs), which suppress anti-tumor or anti-viral immune responses, are expanded in the peripheral blood and tissues of patients/animals with cancer or viral infectious diseases. We here show that in chronic SIV infection of Indian rhesus macaques, the frequency of MDSCs in the bone marrow (BM) was paradoxically and unexpectedly decreased, but increased in peripheral blood. Reduction of BM MDSCs was found in both CD14⁺MDSC and Lin⁻CD15⁺MDSC subsets. The reduction of MDSCs correlated with high plasma viral loads and low CD4⁺ T cell counts, suggesting that depletion of BM MDSCs was associated with SIV/AIDS disease progression. Of note, in SHIV_{SE162P4}-infected macaques, which naturally control viral replication within a few months of infection, the freguency of MDSCs in the bone marrow was unchanged. To investigate the mechanisms by which BM MDSCs were reduced during chronic SIV infection, we tested several hypotheses: depletion due to viral infection, alterations in MDSC trafficking, and/or poor MDSC replenishment. We found that the possible mobilization of MDSCs from BM to peripheral tissues and the slow self-replenishment of MDSCs in the BM, along with the viral infection-induced depletion, all contribute to the observed BM MDSC reduction. We first demonstrate MDSC SIV infection in vivo. Correlation between BM CD14⁺MDSC reduction and CD8⁺ T cell activation in tissues is consistent with decreased immune suppression by MDSCs. Thus, depletion of BM MDSCs may contribute to the pathologic immune activation during chronic SIV infection and by extension HIV infection.

Author summary

Both cancer and infectious diseases including HIV/AIDS lead to the accumulation of myeloid-derived suppressor cells (MDSCs), which can effectively suppress anti-tumor and anti-viral T cell responses to dampen protective immunity. Using a macaque model, we found unexpectedly that the MDSCs in bone marrow (BM) decreased after chronic

simian immunodeficiency virus (SIV) infection compared with healthy controls. This was in sharp contrast to the general increase of MDSCs observed in BM during cancer and other infectious/inflammatory diseases, and also contrary to the MDSC expansion in HIV/SIV-infected PBMCs. We further demonstrated that the loss of MDSCs in the bone marrow was associated with the progression to AIDS disease. Investigating the mechanisms by which the MDSCs were decreased in the SIV-infected bone marrow, we found that the possible mobilization of MDSCs from bone marrow to peripheral tissues and the slow self-replenishment of MDSCs in the bone marrow, along with the viral infectioninduced depletion, all contribute to the observed bone marrow MDSC reduction. Indeed, this is the first demonstration to our knowledge of SIV infection of MDSCs in vivo. Because of the suppressive nature of the MDSCs, the CD8⁺ T cells might not be effective in killing the virally infected MDSCs. It is tempting to speculate that MDSCs may constitute latent reservoirs. Overall, our data showed that MDSCs act as a double-edged sword in HIV/SIV-infection, and the decrease of MDSCs in bone marrow after SIV infection could serve as an indicator of immune regulatory exhaustion and also contribute to the observed immune hyperactivation seen in HIV/AIDS.

Introduction

Myeloid-derived Suppressor Cells (MDSCs) are immature cells of myeloid origin, frequently found in tumor microenvironments and in the blood of cancer patients [1-3]. Bone marrow (BM) is the reservoir for MDSCs. Under homeostatic conditions, there is a delicate balance between the immature MDSCs and matured myeloid cells in the BM. Normally, only matured myeloid cells are released from the BM to peripheral blood and tissues. MDSCs maintain a relatively low level in peripheral blood and tissues, and do not expand under normal physiological conditions. In healthy mice, Gr-1⁺CD11b⁺ MDSCs constitute 20–30% of the total cells in BM, 2–4% of peripheral blood cells, 2–4% of spleen cells, 2–5% of liver cells, and <1% of lymph node cells [2, 4, 5]. Cancer leads to MDSC expansion in tumors, and lymphoid tissues, reaching up to 50% of the total cells in lymph nodes, and bone marrow of tumor-bearing mice [6], and patients with pancreatic cancers [7]. MDSCs were implicated in the suppression of different immune cells including T cells and NK cells[8, 9]. Several suppressive mechanisms of MDSCs have been described including: (1) direct suppressive activities through the production of arginase1(ARG1), indoleamine 2,3-dioxygenase (IDO), reactive oxygen species(ROS), inducible nitric oxide synthase (iNOS), TGF-β, IL-10, and PD-L1 expression; (2) the expansion of regulatory T cells (Treg)[10-16]. In humans, MDSCs are a heterogeneous population without specific markers. Two main subtypes, monocytic, and polymorphonuclear (or granulocytic)-MDSCs, have been described in humans based on their phenotype and functions. Monocytic(M)-MDSCs were defined as HLA-DR^{-/low}CD33⁺CD11b⁺CD14⁺ (hereafter referred to as CD14⁺ MDSC), while polymorphonuclear (PMN, also called granulocytic)-MDSCs were defined as Lin⁻HLA-DR^{-/low}CD33⁺CD11b⁺CD15⁺(hereafter referred to as Lin⁻CD15⁺MDSCs) [8, 17–19]. Recently a new subset called early-stage MDSCs (eMDSCs) was added with phenotypical makers as Lin⁻(including CD3, CD14, CD15, CD19, CD56) HLA-DR^{-/low}CD33⁺ [18]. In some of the studies, cancer leads to the expansion of both PMN and M-MDSCs, with PMN-MDSC as the predominant subset [19, 20]. Both subsets demonstrated a suppressive effect in vitro. However, M-MDSCs have been identified as the more suppressive subset in vivo, and the loss of PMN-MDSC has not altered tumor incidence [21]. MDSC expansion also occurs in non-cancer settings [12, 22–29]. For example, in late septic mice: Gr-1⁺CD11b⁺

MDSCs were increased dramatically with up to 40% of the total cells in spleen (vs. 2–4% in normal spleen), 90% of the total cells in BM (vs 20–30% in normal BM), and 3–4% of the total cells in lymph nodes (vs. <1% in normal lymph nodes)[30, 31].

MDSCs are significantly increased in peripheral blood of HIV/simian immunodeficiency virus (SIV)-infected individuals/macaques [12, 22, 23]. Similar to their roles in the tumor microenvironment, MDSCs in HIV/SIV-infections effectively suppress the function of T cells, and thus play an important role in dampening protective immunity [8, 12, 22, 23]. However, there are no details known regarding MDSC distribution and their roles in different tissues during HIV infection in humans. Because SIV-macaque models recapitulate the pathogenesis of HIV infections, they are considered one of the best animal models for AIDS research. In this study, we investigated MDSC tissue distribution, and possible MDSC roles during SIV infection using single suspension cells obtained from eight different anatomic compartments of SIVmac251-infected rhesus macaques. Surprisingly we found that MDSCs in the BM were paradoxically decreased in chronic SIV infection (dropping from 22% to 8%). Furthermore, the reduction of BM MDSC correlated with high plasma viral loads (VL), low CD4⁺ T cell preservation, and general high CD8⁺ T cell immune activation. We found that an increase in the trafficking and continuous mobilization of the MDSCs from bone marrow to peripheral blood and tissues might be one cause, along with poor replenishment of MDSCs in BM. A second cause is our discovery of viral infection of MDSCs that may further exacerbate the decrease of MDSCs. Taken together, our data suggest that the decrease in BM MDSC may contribute to immune deficiency and immune activation in macaques chronically infected with SIV.

Results

High frequencies of MDSCs in BM, liver, and PBMC, but not in lymph nodes (LN)

Because information on tissue distribution of the MDSCs in normal and HIV-1 infected humans is limited, we investigated the distribution of MDSCs in eight different tissue compartments of SIVmac251-infected Indian rhesus macaques. Single cell suspensions from tissues of 12 SIVmac251 chronically infected macaques (14 months post-infection) were obtained and subjected to flow cytometric analysis of MDSCs [23]. We examined the frequencies of two MDSC subsets, which were both HLA-DR^{-/low}CD33⁺CD11b⁺ and either Lin^{-/low} CD15⁺(Lin⁻CD15⁺MDSC) or CD14⁺ (CD14⁺ MDSC) (Fig 1A) [19, 23, 32–34]. These animals showed a typical dynamic of the plasma VL after pathogenic SIVmac251 infection with peak VL at Week 2 followed by persistent set-point VL from Week 6 to the end study at week 60 (Fig 1B). Among the eight compartments from the chronically infected macaques, interestingly, high frequencies of both types of MDSCs were found in BM (4.8% and 3.1% for Lin⁻CD15⁺ and CD14⁺ MDSCs, respectively, P<0.05 for BM vs. all the other compartments except liver and PBMC in both types of MDSCs after Dunn's multiple comparison tests), liver (0.8% and 2.0% for Lin⁻CD15⁺ and CD14⁺ MDSCs, P<0.05 for liver vs. axillary LN, mesenteric LN, and colon lamina propria (LP) in both types of MDSCs after Dunn's multiple comparison tests), and PBMC (0.3% and 1.1% for Lin⁻ CD15⁺ and CD14⁺ MDSCs, P<0.05 for liver vs. axillary LN, mesenteric LN, and colon LP in both types of MDSCs after Dunn's multiple comparison tests) with intermediate levels of MDSCs in spleen and gut; but very few of either subset in the LNs (Fig 1C and 1D). For both subsets of MDSCs, when we analyzed the associations among different tissues, we could not find any correlations between any pairs of compartments (S2 and S3 Tables), suggesting compartmentalization. For example, the animal with the highest CD14⁺MDSCs in the BM (8%) only had 0.2% of CD14⁺MDSCs in the PBMC.











Fig 1. High frequencies of MDSCs were found in bone marrow, liver, and PBMC, but very few in the LNs of the chronically SIVinfected macaques. (A) The gating strategy of CD14⁺MDSCs and Lin⁻ CD15⁺MDSCs from the SIV-infected bone marrow. The same gating strategy was used for other compartments. (B) The kinetics of plasma VLs in the 12 SIV_{mac251}-infected macaques showed persistent viral infection (mean±SEM). Single suspension cells from eight compartments of 12 SIVmac251 chronically (14 months-) infected Indian rhesus macaques were used to measure the frequencies of CD14⁺ MDSCs (C), Lin⁻ CD15⁺MDSCs (D), gag-specific CM9 Dextramer⁺ (E), and Ki67⁺ (F) in CD8⁺ T cells. In C-D, P<0.05 for bone marrow vs. all the other compartments except liver and PBMC; P<0.05 for liver vs. axillary LN, mesenteric LN, and colon LP; P<0.05 for PBMC vs. axillary LN, mesenteric LN, spleen and bone marrow after Dunn's multiple comparison tests.

https://doi.org/10.1371/journal.ppat.1006395.g001

In addition to MDSCs, we also measured viral-specific CD8⁺ T cell responses by gag CM9 peptide-Mamu A*01 Dextramer as well as the total proliferating CD8⁺ T cells by Ki67 in these animals, because both can be affected by MDSCs. Because all the animals were Mamu A*01⁺, SIVgag-CM9 Dextramer⁺ cell frequencies were measured to represent viral-specific CD8⁺ T cell responses[35]. As shown in Fig 1E, SIVgag-CM9 Dextramer⁺ cell frequencies in PBMC was 1.7%, much lower than in the remaining tissues, which ranged from 3% to 5% of total $CD8^+T$ cells (P<0.05 for PBMC vs. axillary LN, mesenteric LN, spleen and BM after Dunn's multiple comparison tests). However, the frequencies of CM9 Dextramer⁺ in total CD8⁺ T cells from different tissues were highly correlated with each other (S4 Table), suggesting the equilibration of antigen-specific T cell responses among different tissue compartments. The proportion of total CD8⁺ T cells that expressed Ki67 ranged from 35 to 54%, without significant differences among the compartments (Fig 1F). We found that the frequencies of Ki67⁺ within the total CD8⁺ T cells from different tissues were also associated with each other (S5 Table). Overall, the tissue distribution patterns of viral-specific CM9 Dextramer⁺ and total Ki67⁺ CD8⁺ T cells were different from those of MDSCs in the same SIV-infected macaques. It is of note that the MDSCs did not accumulate in the lymphoid tissues of the SIV-infected macaques during the chronic stage, which marked a sharp contrast to cancer and some infectious diseases[36].

MDSCs in bone marrow were decreased after SIV infection, and correlated with markers of SIV disease progression

Due to the tissue distribution shown in Fig 1, we then focused on the MDSCs in BM, liver and PBMC. One important question was whether the frequencies of the MDSC subsets were altered during chronic SIV infections. To make the comparisons, we measured the frequencies of MDSCs among PBMCs (from two cohorts, n = 40 and 28), liver (from two cohorts, n = 4and 4) and BM (from two cohorts, n = 28 and 12) from naïve macaques. Using a similar gating strategy (Fig 2A), we found that while naïve animals had a low frequency of MDSCs in PBMCs (0.09±0.02% and 0.20±0.04% for CD14⁺ and Lin⁻CD15⁺MDSC), PBMC MDSC frequencies were elevated after SIV infection (1.15±0.14% and 0.27±0.08% for CD14⁺ and Lin⁻CD15⁺ MDSC in PBMCs) (Fig 2B). This is consistent with the results from previous studies showing an expansion of MDSCs in the PBMCs after HIV-1/SIV infection[12, 22, 23]. Though both subsets of MDSCs were more prevalent, we observed a more dramatic increase of the CD14⁺ MDSC subset than the Lin⁻CD15⁺MDSC subset: 13-fold increase of CD14⁺ MDSC compared with almost no change of Lin⁻CD15⁺MDSC after SIV chronic infections. Due to the small number of naïve liver samples that were available, we could not reach any conclusions as to effect of SIV infection on liver MDSCs (Fig 2C). Surprisingly, we found that the total number of MDSCs (defined as the sum of CD14⁺ MDSC and Lin⁻CD15⁺MDSC) in BM was paradoxically decreased 14 months after SIVmac251 infection (Fig 2D). Specifically, there was roughly a 4.7-fold reduction in the CD14⁺ MDSC subset (from 14.5±1.3% in naïve animals vs. 3.1 ±0.07% in SIV-infected ones, P<0.0001), and a 1.6-fold reduction in the Lin⁻CD15⁺MDSC subset $(7.7\pm0.6 \text{ vs. } 4.8\pm0.7\% \text{ respectively}, P = 0.005)$ (Fig 2D). This pattern of the MDSC



Fig 2. CD14⁺ and Lin⁻**CD15⁺MDSCs** were decreased in BM and increased in peripheral blood during chronic SIV infection. (A) The gating strategy of CD14⁺MDSCs and Lin⁻CD15⁺ MDSCs from the naïve bone marrow. (B-D) The MDSC frequencies of the naïve macaques in the PBMCs (68 naïve animals, the purple and orange squares denote the PBMC samples from two different cohorts of naïve animals.), liver (8 animals; purple and blue squares represent livers from 2 different cohorts of naïve animals) and bone marrow (40 naïve animals, the purple and orange squares from two different cohorts of naïve animals.) were measured and compared with those of the SIV-infected macaques (n = 12 for each compartments). Mann-Whitney was used for the comparisons.

https://doi.org/10.1371/journal.ppat.1006395.g002

distribution in the SIV-infected animals was markedly different from that in tumor-bearing mice, where BM MDSCs are elevated [4, 5]. Furthermore, we found two subpopulations of CD14⁺ MDSCs in the BM: CD14^{high} and CD14^{intermediate} MDSCs (S1 Fig). Most CD14^{intermediate} population were HLA-DR^{-/low} cells in SIV-infected and naïve BM. This HLA-DR^{-/low} CD14^{intermediate} population was prominent in the naive animals, but lower in the SIV-infected animals (S1 Fig). Compared to naïve BM, the CD14⁺ population was also decreased in the SIV-infected BM (S2 Fig).

Because MDSCs, especially Lin⁻CD15⁺ subset, are sensitive to cryopreservation [37], and we used frozen cells throughout the whole study, there is a possibility that SIV infection might differentially affect the loss of SIV-infected MDSCs compared to naïve cells. To test this, we compared the frequencies of MDSC subsets in paired specimens of fresh vs frozen BM and PBMC obtained from the SIV-infected (BM: n = 6; PBMC: n = 7) and naïve animals (BM: n = 12; PBMC: n = 5). The PBMC and BM samples from the SIV-infected and naïve animals were isolated, and divided into two aliquots, one was immediately stained and analyzed by flow-cytometry, the other was cryopreserved in liquid N₂ and thawed for testing 4–5 months later. After cryopreservation and thawing, 70–80% of CD14⁺MDSCs were preserved in the naïve and SIV-infected BM samples; 65% of CD14⁺MDSCs were also preserved in the SIVinfected PBMC samples (S3 Fig). Consistent with the previous study [37], only 20% of Lin⁻CD15⁺ MDSCs were preserved in the naïve and SIV-infected BM samples (S3 Fig), suggesting Lin⁻CD15⁺MDSC subset was more sensitive to cryopreservation. Importantly, we did not observe any significant difference of MDSC subset loss for the BM samples between the SIV-infected and naïve animals. Thus, SIV infection does not differentially affect the loss of MDSC subsets from BM compared to naïve ones.

We then asked whether SIV VLs correlated with the reduction of BM MDSCs. Among the 12 SIV-infected macaques, five of them controlled SIV replication, with plasma VLs below 5,000 copies/ml. The remaining 7 animals did not control SIV replication, with VLs >5,000 copies/ml. We found that the controllers maintained high BM MDSC frequencies, whereas the non-controllers had reduced BM MDSC frequencies (Fig 3A). This change was not found in the PBMC or the liver, although the liver showed the same trend (Fig 3A). Indeed, the frequencies of both subsets of MDSCs in BM inversely correlated with VL (Fig 3B). In the BM of chronically SIV-infected macaques, the frequencies of CD14^{intermediate} MDSCs inversely correlated with plasma viral loads better than that of CD14^{high} MDSCs (S4 Fig). We also found that both subsets of MDSCs in the BM positively correlated with absolute blood CD4 count ratio (necropsy end-point CD4⁺ T cell count/pre-challenge CD4⁺ T cell counts suggests that MDSC depletion may contribute to disease progression, or at least serve as marker of disease progression.

SHIV_{SF162P4}-infected macaques, which had undetectable plasma VL 11 months post-infection, maintained their normal frequency of MDSCs in bone marrow

The data suggested that exposure to pathogenic SIV infection decreased MDSC frequencies in the BM. We then asked if similar changes occurred in Indian rhesus macaques infected with



Fig 3. After chronic SIV-infection, non-controller (log VL>3.7) had lower MDSC in the bone marrow than those of the controllers (log VL<3.7). (A) Among the 12 SIV-infected macaques, 5 of them were controllers with plasma viral loads below 5,000 copies/ml, while the remainder of them (n = 7) were non-controllers with viral loads greater than 5,000 copies/ml. MDSC (both CD14⁺ MDSC and Lin⁻CD15⁺ MDSC subsets) in the bone marrow of the controllers maintained relatively higher levels compared with those of the non-controllers. Mann-Whitney was used for the comparisons. (B) The MDSCs in the bone marrow inversely correlated with the plasma VLs, and positively correlated with absolute CD4 count ratio (Necropsy CD4⁺ T cell count/pre-challenge CD4⁺ T cell count). Spearman analysis was used for correlations. Total MDSC was the sum of CD14⁺MDSC and Lin⁻CD15⁺ MDSC subsets.

https://doi.org/10.1371/journal.ppat.1006395.g003

SHIV_{SF162P4}. All the SHIV_{SF162P4}-infected animals demonstrated high peak VLs during the first 6 weeks of infection, followed by viral control 3 to 4 months post-infection (Fig 4A). At 11 months post-infection, when VL was near or below the detection limit, there was no change in BM MDSC frequency (Fig 4B). In those samples, there was no change in either the Lin⁻CD15⁺ MDSC or CD14⁺ MDSC (Fig 4B). The same was also true for MDSCs in the 11-month post SHIV-exposed PBMC and liver (Fig 4C and 4D). Thus, infection with SHIV without persistent viremia did not affect MDSC frequency in BM, but infection with pathogenic SIV depleted BM MDSC.

Mechanisms of MDSC reduction in the bone marrow of macaques chronically infected with SIV

We next investigated what caused the reduction of MDSCs especially in the CD14⁺ subset in the BM of the SIV-infected macaques. We proposed three hypotheses: depletion due to viral infection, increase in MDSC mobilization/trafficking out of BM, and poor MDSC replenishment from hematopoietic stem cells.

MDSC infection by SIV in vivo. The SHIV infection data suggest that only pathogenic viruses can decrease MDSC frequency. Thus, the MDSC decrease may be due in part to direct infection with SIV and virus-mediated cell death. In fact, HIV-1 infects MDSCs in vitro [22]. We asked whether MDSCs were infected by SIV in vivo, which had not been described previously. Using FACS, we purified MDSCs from the PBMCs or spleens of the chronically SIVinfected macaques (Fig 5A, purity>95%). The isolated RNAs from purified MDSCs were then subjected to qPCR to detect the expression levels of SIV gag, CD3 and CD4, while purified CD4⁺ T cells were used as positive controls. We found that all the MDSC samples were negative for CD3 mRNA, suggesting no T cell contamination for the MDSC subsets. Most of the MDSC samples (4 of 6) were positive for SIV gag and CD4, indicating infection of the MDSCs by SIVmac251. Importantly, the ratio of SIVgag to CD4 was much higher in the MDSC population than the CD4⁺ T cell population, further confirming that the virus detected could not be due to contaminating CD4⁺ T cells (Fig 5B, S5 Fig). Concordant with that, we also found that the SIVgag expression level is comparable between CD4⁺ T cells and MDSCs (ratio around 1), whereas the CD4 expression level was much lower in MDSCs than in the CD4⁺ T cell subset (Fig 5C, S6 Fig), further supporting the conclusion that MDSC-associated viral RNA could not be due to CD4⁺ T cell contamination. Since the SIVgag assay cannot distinguish the genomic RNA in virions or infected cells from the SIVgag mRNA, we performed real-time RT-PCR assays to detect the spliced viral mRNAs that are produced only in infected cells. Spliced viral mRNAs can thus act as a marker of active virus replication. While all the T cell samples were positive for spliced viral mRNA, out of the four MDSC samples that were positive for SIVgag, we found that 2 of them were also positive for spliced viral mRNA (S7 Fig). Given the importance of this finding, we tested additional animals in an independent experiment to confirm the conclusion of productive infection of MDSCs. We performed another MDSC sorting using six SIVmac251-infected BM samples to detect the spliced SIV mRNA. The sorted CD4⁺T cells















Lin-CD15+ MDSC-PBMC





Lin-CD15+MDSC in liver



https://doi.org/10.1371/journal.ppat.1006395.g004





were 10-fold serially diluted to generate a titration curve so that the detection limit of spliced SIV /CD3 mRNA was from 90 to 900000 CD4⁺T cells. Four sorted MDSC samples were negative for CD3 mRNA, and three out of these four were positive for spliced SIV mRNA (<u>S7 Fig</u>). Overall, these two independent experiments (<u>S7 Fig</u>) demonstrated that the spliced SIV mRNA were present in the MDSC (5 out of 8 CD3 negative) samples, indicating productive infection rather than BM-phagocytosis of infected cells.

To further confirm that SIV infects MDSC in-vivo, we sorted MDSCs and CD4⁺ T cells from the blood of 6 additional SIV-infected animals. We isolated RNA from MDSC and CD4⁺ T cell samples and performed 10-fold serial dilutions of CD4⁺ T cells (up to 5 dilutions) to generate titration curves, which were linear on a log-log scale (black dots in Fig 5D). After RNA isolation and cDNA synthesis, each of the CD4⁺ T and MDSC samples was subjected to qPCR analysis for detection of SIVgag, CD3, CD4, and GAPDH using the same amounts of cDNAs (red symbols superimposed on titration curves represent MDSCs from the same animal on the same scale). Out of these 6 animals, we failed to isolate RNA from one sample due to a low number of MDSC sorted. Out of the five samples from which we have successfully isolated RNA, we could not detect SIVgag from one sorted MDSC sample (the detection limitation for Gag is 50 copies/well). Among the four MDSC samples scoring positive for SIVgag, we detected CD3 mRNA from two of them (NHP-a and -b) (Fig 5D), indicating possible contamination or phagocytosis of the T cells as suggested by another study[38]. However, SIV gag was detected in the two remaining SIVgag-positive samples (NHP-c and-d), that were negative for CD3 mRNA, indicating infection of MDSC. The sensitivity to detect T cell contamination in these two animals was 27 and 80 CD4⁺ T cells (S7 Fig). Note that the housekeeping gene GAPDH was expressed at the same level per cell in the MDSCs (red) as in the CD4⁺ T cells (black dots) (S7 Fig).

We also assessed SIVgag p27 expression in BM cells of SIVmac251 chronically infected BM by Immunocytochemistry. In all 6 SIV⁺ animals a variable number of Lin⁻CD15⁺ and CD14⁺ MDSCs were positive for SIV P27. Thus, both subsets of MDSC are infected by SIVmac251 invivo (Fig 6). Results from one representative experiment are shown (Fig 6A). The summary of three independent experiments demonstrated that the P27⁺% in both subsets of MDSCs from the SIV-infected bone marrow was significantly higher than in naïve animals (Fig 6B, S8 Fig). Note that the MDSCs from SIV-infected animals not only had substantially more events in the positive gate, but also had brighter staining indicating real infection. Thus, SIV infects both subsets of MDSCs *in vivo* and this could directly contribute to their depletion from the BM of infected animals.

MDSC trafficking from BM to blood. We then explored the MDSC mobilization/trafficking hypothesis, in which MDSCs moved from BM to blood and peripheral tissues with an accelerated rate after SIV-infection compared with those of the uninfected controls.

To test this hypothesis, we performed *in vitro* chemotaxis assays to measure the movement of BM single-cell suspensions towards the plasma of pre- and post-infected animals. We observed that the MDSC chemotaxis index toward the plasma of the post-infected samples was significantly higher than those of the pre-infected plasma (Fig 7A). However, the chemotaxis indexes of total cells and of T cells did not change after SIV-infection (Fig 7A). These data suggested a preferential trafficking of MDSCs, but not T cells or total cells, from BM to blood after SIV-infection. We then investigated whether the MDSC chemotaxis index was associated with the frequencies of MDSC in the SIV-infected PBMC. We observed trends of positive correlations between the MDSC chemotaxis index from the SIV-infected plasma and the frequencies of CD14⁺, CD15⁺Lin⁻, and total MDSCs in SIV-infected PBMCs, suggesting the important role of MDSC trafficking from bone marrow to blood during SIV infection (Fig 7B).

We then sought to characterize the chemokine(s) responsible for the trafficking of MDSCs. During cancer and HIV infection, the balance between MDSCs in BM and peripheral tissues is interrupted due to the massive cytokine/chemokine storm that characterizes these conditions. CCL2 (also known as MCP-1) is a key chemokine involved in MDSC trafficking. In tumor-bearing mice, the CCR2-receptor is highly expressed by most MDSC subsets, and CCL2 alone is sufficient to induce migration of MDSCs [39]. It has been shown that CCL2 is up-regulated upon HIV infection [40]. Based on this, we proposed that changes in CCL2 levels might be



Fig 6. Flow cytometric detection of gag-P27⁺ MDSCs and CD4⁺T cells in the bone marrow of SIV-infected or naive animals. (A) One representative P27 staining from one experiment with 7 naïve and 3 SIV-infected animals (B) Summary of three independent experiments with total 6 SIV-infected, and 12 naïve animals. The SIV-infected and naïve animals from the same experiment were shown in same color. Each data point represents one individual animal. The Man-Whitney tests were used for comparisons.

responsible for the trafficking of the MDSCs from BM to the periphery in the SIV-infected macaques. Indeed, we found that the expression level of CCL2 in the plasma of SIV chronically infected macaques was significantly greater than that of the naïve control samples (Fig 7C). Furthermore, it was also positively correlated with the CD14⁺ MDSC subset (Fig 7C), but not the Lin⁻CD15⁺MDSC (Spearman P = 0.9) in the PBMCs, suggesting that CCL2 might be responsible for the accumulation of the CD14⁺ MDSC in the peripheral blood. However, we did not find a correlation between CCL2 and the MDSC chemotaxis index measured in *in vitro* chemotaxis assays, suggesting that other chemokines, besides CCL2, in the SIV-infected plasma, play roles in determining the migration of MDSCs from BM to blood (Fig 7C).

MDSC replenishment from hematopoietic stem cells. Another important factor that could affect the frequency of the MDSCs in the bone marrow is rate of MDSC replenishment. Due to the infection of the BM accessory cells, impaired stromal function and alteration of the hematopoietic growth factor network was frequently observed in the BM of patients with HIV-1 infection[41–43]. These events directly lead to disruption of normal hematopoiesis in



Fig 7. Preferential trafficking of MDSCs from BM to peripheral blood during chronic SIV infection measured by in vitro chemotaxis assays. (A) Chemotaxis index of MDSCs, T cells and total cells were compared before and after SIV infections. Wilcoxon tests were used for comparisons. (B) Correlation trends were observed between MDSC chemotaxis index and CD14⁺ or Lin⁻ CD15⁺ or total MDSCs in the PBMCs after chronic SIV infection. Spearman analysis was used for correlations. (C) Plasma CCL2 level was higher in the SIV-infected macaques than in naïve ones, and correlated with CD14⁺ MDSC in the PBMCs, but did not correlate with the MDSC chemotaxis index. Man-Whitney tests were used for comparisons, and Spearman analysis was used for correlations.

the BM[44, 45]. Our hypothesis was that chronic SIVmac251 infection might impair the proliferation of MDSCs in the BM. We measured the cell proliferation marker Ki67 to assess the proliferative replenishment rate of MDSCs in the BM. Indeed, proliferation of both CD14⁺ and Lin⁻CD15⁺MDSCs in BM was lower in SIV-infected macaques than in naïve controls (Fig 8A–8C). In the SHIV-infected macaques, which spontaneously controlled their viral replication several months after the infection, we found that the proliferation of the CD14⁺ and Lin⁻CD15⁺ MDSCs in the bone marrow 11 months post-infection were also significantly lower than those of the uninfected controls (Fig 8A–8C). Even through the proliferation of both subsets of MDSCs were decreased in SHIV chronically infected BM, the frequency of both subsets were maintained at pre-infection levels (Fig 4B). Interestingly, in both SIV-and SHIV-infected animals, we observed a bimodal Ki67-positive population



Fig 8. MDSC proliferation was decreased in the bone marrow of SIV-infected macaques. (A) Representative Ki67⁺% of the gated CD14⁺ and Lin⁻ CD15⁺MDSCs in the bone marrow from naïve, SIV- or SHIV-infected macaques. (B, C) Ki67⁺% of the gated CD14⁺ and Lin⁻ CD15⁺MDSCs in the bone marrow was lower in the SIV-infected macaques than in the naïve, or SHIV-infected animals. Man-Whitney were used for comparisons between the naïve and SIV- or SHIV-infected macaques. The black and orange dots denote the bone marrow samples from two different cohorts of naïve animals.

Ki67^{high} subsets in CD14⁺ and Lin⁻CD15⁺MDSCs, which were almost absent from those of the uninfected controls (Fig 8A).

Thus, the slow self-replenishment of BM MDSCs probably exacerbated the decrease of MDSCs when they were killed by virus, or migrated out of the BM.

MDSC reduction in bone marrow was associated with increased T cell immune activation

We then sought to clarify the consequences of the MDSC reduction in the BM. Because the main known function of MDSCs is to limit excessive T cell proliferation, and thus control tissue damage, we proposed that BM MDSC reduction would be associated with T cell activation, especially among $CD8^+$ T cells. We measured the frequencies of Ki67⁺ CD8⁺ T cells in the

tissues of the SIV-infected macaques, and then performed correlation analyses between frequency of BM CD14⁺ / Lin⁻CD15⁺MDSCs and the frequencies of Ki67⁺ CD8⁺ T cells in these compartments. We found that the frequency of BM CD14⁺ MDSCs inversely correlated with the frequency of Ki67⁺CD8⁺ T cells in BM, PBMC, spleen, ALN, but not MLN, liver, ileum LP and colonic LP (although MLN and colonic LP showed nearly significant trends) (Fig 9A–9H); whereas Lin⁻CD15⁺ MDSC did not correlate with the frequency of Ki67⁺CD8⁺ T cells in any of the compartments except PBMC (S9 Fig). Overall, the association of BM CD14⁺ MDSC frequency with CD8⁺ T cell immune activation, plasma vRNA levels, and CD4⁺ T cell loss suggests that the reduction in BM CD14⁺ MDSC contributes to the pathogenesis of the HIV/SIV infections.

Discussion

Cancer and some infectious/inflammatory diseases lead to MDSC accumulation in peripheral blood and multiple tissue compartments, including BM[1-6]. A higher frequency of MDSCs in a cancer setting is usually associated with a poor prognosis [7, 34]. In this study, we found unexpectedly that BM MDSCs paradoxically decreased after chronic SIV infection compared with healthy controls. This was in sharp contrast to the general increase of MDSCs observed in BM during cancer and other infectious/inflammatory diseases, and also contrary to the MDSC expansion in HIV/SIV-infected PBMCs [12, 22, 23]. We further demonstrated that even though the changes were among both CD14⁺ and Lin⁻CD15⁺MDSC subsets in BM, and both subsets correlated with high plasma vRNA level and CD4⁺ T cell loss, it was only the reduction of CD14⁺ MDSC that correlated with the high levels of CD8⁺ T cell activation. These changes suggest that the loss of BM MDSCs, especially the CD14⁺ subset, was associated with the progression to HIV/AIDS disease. We then investigated the alteration of MDSCs in the BM of a less pathogenic SHIV-macaque model, in which the animals had moderate plasma vRNA levels at the first 6 weeks of infection, but spontaneously controlled the viral replication after several months. These animals had no signs of immune activation, and did not develop AIDSrelated disease. Indeed, in these animals, we found that both subsets of BM MDSCs were unchanged 11 months post-infection. This further supported the association of MDSC decrease in BM with HIV/AIDS disease progression.

During HIV/SIV infection, MDSCs are thought to contribute to immune-pathogenesis by dampening protective immunity through the direct inhibition of T cell function, especially the viral-specific CD8⁺ T cell responses, and thus are deleterious [12, 22, 23]. However, our data suggest more complex and somehow paradoxical roles of MDSCs in HIV/SIV infections. MDSCs suppress anti-viral specific immune responses, which hindered the control of HIV replication [22, 23]. However, MDSCs also have the capacity to antagonize immune activation, which plays a key role in driving viral transmission and replication and many of the disease processes associated with AIDS [46-48]. Consistent with this, the CD14⁺ MDSCs in the BM inversely correlated with immune activation (Ki67⁺CD8⁺T cells) of multiple tissue compartments, suggesting that depletion of BM MDSCs contributes to the immune activation in chronically infected macaques. During late HIV infection, the balance of immune suppression and immune activation tips in favor of the latter due to factors such as gut microbe translocations [49, 50]. It has been proposed that MDSCs are responsible for restoring the balance through various mechanisms, including arginase-1, inducible nitric oxide synthase, reactive oxygen species, and induction of regulatory T cells [46, 51-53]. However, the reduction of MDSCs in the BM may impair this function, and contribute to the relentless immune activation in HIV/SIV chronically infected patients/animals.



Fig 9. Decreased CD14⁺ MDSCs in bone marrow were associated with increased immune activation. The frequency of CD14⁺ MDSCs was inversely correlated with the Ki67⁺ CD8⁺ T cell frequencies in the SIV-infected bone marrow (A), PBMC (B), spleen (D), Axillary LN (E) and Mesenteric LN (F), but not liver (C), ileum LP (G) or colon LP (H). (although MLN and colon LP showed strong trends (p = 0.06). Each data point represents one animal. Spearman analysis was used for correlations.

We are puzzled by the fact that the MDSC reduction in BM correlated with the CD8⁺ T cell proliferation in not only the MDSC-high tissues but also the MDSC-low tissues. MDSCs did not play a direct role in inhibiting CD8⁺ T cells proliferation in situ in the tissues with low MDSC frequencies, whereas in the MDSC-high compartments, MDSCs could inhibit the proliferation of CD8⁺ T cells efficiently. When the CD8⁺ T cells are equilibrated well among different tissue compartments in the SIV-infected animals, MDSCs suppressing T cell in high-MDSC tissues could affect the levels of T cells in the MDSC-low tissues indirectly when they re-equilibrate. MDSCs could play an important role in suppressing immune activation during HIV infection. Another suppressive mechanism of MDSCs could be the induction of CD4⁺ Treg cells, which could travel to other MDSC-low compartments, and suppress immune responses locally[12–14]. Furthermore, MDSCs induced a type 2 polarization of macrophages in tissues, which were functionally inhibitory [54, 55].

It is worth mentioning that even though both subsets of MDSCs were decreased in the chronically SIV-infected BM, and associated with markers of disease progression: high plasma vRNA levels, low CD4⁺ T cell counts, only loss of CD14⁺ MDSCs was associated with high immune activation. Thus CD14⁺ MDSC and Lin⁻ CD15⁺ MDSC were not only phenotypically distinct, but also played different roles in SIV infections. In cancer settings, CD14⁺ MDSCs have been shown to be the dominant suppressive populations of MDSCs, and correlated with cancer incidence, whereas G-MDSC did not correlate with clinical outcomes, even though both subsets were increased[21]. We also observed more dramatic changes of CD14⁺ MDSC subsets in both PBMC and BM after SIV infection. This implied that the two subsets of MDSCs were under different regulatory mechanisms, and CD14⁺ MDSCs might be the main subset that respond to SIV infection, and determine the outcome of SIV infections. However, as cryopreservation changes the PMN-MDSC (Lin⁻CD15⁺) numbers and function[37, 55], and all the samples used in this study were cryopreserved ones, we should be cautious on the interpretation of data on this subset.

In the present study, to investigate the mechanisms by which the MDSCs were decreased in the SIV-infected BM, we found that infection by virus, migration out of BM, and/or poor self-replenishment of MDSCs might all play important roles. With the data we have, it is impossible to dissect which were more critical. It has been reported that infectious/inactivated HIV, or gp120/gp41, and tat proteins directly promoted MDSC expansion in an *in vitro* culture system using healthy PBMCs [13, 14, 22]. However, we doubt that virus/viral proteins were the key contributors to the MDSC accumulation in the PBMCs of the infected macaques based on 1) dramatic increase of M-MDSCs in the PBMCs of HIV-1-infected patients on antiretroviral therapy with undetectable viremia [14]; 2) no correlations of VLs with MDSC accumulation in PBMC at different time-points in the same animal[23]; and 3) decreased rather than increased MDSCs in BM during chronic SIV infection.

Consistent with the finding of CD14⁺ MDSC susceptibility to HIV-1 infection *in vitro* [22], we further demonstrated by three independent mutually corroborative experimental criteria (gag expression relative to CD3 and CD4 expression, spliced vRNA expression, and p27 protein production) that both subsets of MDSCs were able to be infected *in vivo*. In contrast to the suggestion that the infection of myeloid cells might cause developmental defects in myeloid cells, and result in accumulation of MDSCs in the *in vitro* study [22], our data suggest that infection of MDSCs by SIV *in vivo* might eventually lead to MDSC depletion directly or indirectly. In the SIV-infected PBMC and BM, the SIVgag⁺ MDSCs were a small fraction of the total MDSCs. If infection itself led to MDSC expansion, we would expect a big pool of SIVgag⁺ MDSCs in the infected animals. We noticed that the relative expression level of SIVgag mRNA per cell was comparable between CD4⁺ T cells and MDSCs (Fig 5C), which suggested that MDSCs could be productively infected by SIV, like their CD4⁺ T cell counterparts. Productive

infection was confirmed both by detection of spliced SIV mRNA (S7 Fig), and by staining for expression of viral gag protein (Fig 6), under conditions in which contamination with T cells could be ruled out as an explanation. If MDSCs are productively infected with SIV, viral-induced depletion might play an important role. However, further evidence is needed to confirm whether SIV infection leads to MDSC death.

HIV-1 infected patients usually develop hematopoietic dysfunction, which is mainly mediated by the altered production of stromal cell-derived hematopoietic growth factors, infection of the BM accessory cells, and impaired stromal functionality[41–43, 56]. This abnormality could lead to defects of MDSC replenishment. Indeed, we found decreased proliferation (Ki67⁺%) of both subsets of MDSCs in BM after SIV infection.

The inverse or reciprocal relationship of MDSCs in BM and PBMC, e.g., increased in PBMC and decreased in BM, promoted us to investigate the possibility of MDSC redistribution during SIV infection. We compared the chemotactic activities of plasma from the preand post-SIV infection blood using in vitro chemotaxis assays. It turned out that the chemotaxis index of MDSCs, but not of T cells or total cells, was increased during chronic SIV infection. We believe that this could directly lead to MDSC loss in BM, as a consequence of more MDSCs' migrating out of the BM. The preferential trafficking of MDSCs towards the postinfection plasma suggested that certain chemokine(s)/cytokine(s) in the chronic SIV infected plasma could preferentially pull the MDSCs out of BM. In the attempt to dissect which chemokine(s)/cytokine(s) were responsible, we tested CCL2, known to affect myeloid cell migration. We found that CCL2 correlated with CD14⁺ MDSC accumulation in the PBMC, but did not correlate with MDSC chemotaxis index measured in vitro. Since CCL2 was one of multiple cytokines/chemokines induced by SIV, other chemokines/cytokines might also play roles in attracting MDSCs to the circulation. Antibody blocking for CCL2 would be a good way to assess the role of CCL2 played in chemotaxis. Nevertheless, as MDSCs are such heterogeneous populations and no single chemokine receptor is truly specific for MDSCs, the net chemoattractant effects of multiple chemokines in the plasma might account for the results of the *in* vitro assays.

One caveat was that this was a cross-sectional study of MDSC alteration in BM during the chronic infection stage. A longitudinal study with multiple time-points during acute and chronic infection stages will give us a better picture of the MDSC dynamics, and better understanding of the roles of MDSCs during SIV infection. We speculate that there is a delicate balance of immunity and suppression established at the onset of the SIV infection. During the early stage of the pathogenic SIV infection, it is possible that the MDSC increase/expansion also occurred in BM, as we have observed in the SHIV-infected macaques, in which MDSCs initially significantly increased 4 months post-infection (S10 Fig), and then reverted to normal. In the pathogenic SIV infections, however, the presence of persistent high viral loads, and the poor self-replenishment by MDSC proliferation due to bone marrow stromal functional defects could lead to the gradual decrease of MDSCs in BM. Future studies on the kinetics of MDSCs in the BM of pathogenic SIV infections will clarify this. This depletion of MDSCs eliminates one of the body's key mechanisms to limit tissue damage from aberrant immune activation. Thus, in the setting of acute SIV infection, MDSCs can limit antiviral T cell responses, whereas in chronic infection, BM MDSC depletion can contribute to the disease progression. Overall, MDSCs act as a double-edged sword in HIV/SIV-infection, and the decrease of MDSCs in BM after SIV infection could serve as an indicator of immune regulatory exhaustion. Further elucidating the roles of MDSCs in HIV/SIV infection will contribute to our understanding of the immune-pathogenesis of HIV infections and contribute to the development of therapeutic approaches for HIV/AIDS.

Materials and methods

1. Ethics statement

All the adult Indian rhesus macaques (Macaca mulatta) were used with the approval of Institutional Animal Care and Use Committees. The macaques were housed at the NCI Animal Facility, Bethesda, MD (Protocols No. VB010 and VB011 approved by the NCI IACUC), which is accredited by the Association for Assessment and Accreditation of Laboratory Animal Care (AAALAC) International. The housing and the standard practices closely followed the recommendations of the Standards and Guide for the Care and Use of Laboratory Animals of the United States—National Institutes of Health. All efforts including provision of peri-operative and post-operative analgesia were made to minimize discomfort of the animals. Details of animal welfare, including housing, feeding, environmental enrichment, and steps to minimize suffering, were in accordance with the Guide and the recommendations of the Weatherall report, "The use of non-human primates in research", as approved by the IACUCs. Macaques were housed in temperature controlled facilities with temperature of 21–26°C, humidity of 30%-70%, and a 12 h light/dark cycle. A commercial primate diet and fresh fruit were provided twice daily with water freely available at all times. Macaques were monitored twice daily for overall health including activity, food and water intake. The macaques were singly housed in stainless steel wire-bottomed cages due to the nature of the experiment. Rotating toys, visual and auditory stimuli, and foraging opportunities were provided daily. The animals were anesthetized with approximately 10 mg/kg of ketamine hydrochloride injected intramuscularly for blood and bone marrow collections. When IACUC defined endpoints were reached, macaques were humanely euthanized with an overdose of barbiturate in accordance with the recommendations of the most recent American Veterinary Medical Association Panel on Euthanasia.

2. Animals, viral challenge, and viral load (VL) measurements

The 12 simian immunodeficiency viruses (SIV) mac251-infected animals have been described in our previous studies [23, 57] (S1 Table, cohort 1A). Briefly, the animals were intrarectally challenged by three serial SIVmac251 viruses at 2-week intervals until they have been infected. The SIVmac251 viral stock was provided by Nancy Miller of the National Institute of Allergy and Infectious Diseases (NIAID). Twelve simian-human immunodeficiency virus (SHIV)infected macaques were also included in this study (6 males and 6 females; age: 3,3±1.3 years; weight: 4.6 ± 1.7 kg) (S1 Table, cohort 2A). The animals were intrarectally exposed to 8 serial SHIVSF162P4 viral challenges with a week of interval (1:35 dilution) until they were infected. After the viral challenges, SIV/SHIV RNA levels were monitored by Advanced BioScience Laboratories, Inc. The cut-off threshold for viral RNA detection was 50 copies/ml. Blood and tissue samples from naïve macaques were collected and included in the study for comparisons (S1 Table, cohort 1, 2, 3 and other). Liver tissues from 8 naive (4 adenovirus-exposed, 4 nonexposed, all SIV/SIHV-negative) macaques were used as normal controls (S1 Table, other). Other naïve macaques which have been enrolled in this study were described in <u>S1 Table</u>. Additional SIVmac251 infected bone marrow samples were obtained from California National Primate Research Center (S1 Table).

Preparation of single-cell suspensions from the tissues

Single-cell suspensions from blood, bone marrow, lymph nodes, and spleens were prepared as previously described [58]. We have used the same protocol to process bone marrow and blood samples. Briefly, the samples were first 1:1 diluted with 1XPBS or HBSS, and then the diluted samples were overlaid on lymphocyte separation medium Lympholyte-H (Human, Cedarlane,

Ontario, Canada). After centrifugation at 800 x g for 20 min at room temperature (25°C), with centrifuge brake on OFF position, the middle opaque fluid containing the PBMC/BM cells was collected. The cells were washed with R10 medium three times before cryopreservation in liquid nitrogen freezers. Ileum lamina propria, colon lamina propria, and liver were collected from the macaques as previously described with modification[23, 57–59]. Briefly, the tissues were first rinsed 2–3 times with 1XHBSS containing 5mM EDTA and 1-2mM DTT to remove mucous and intestinal intraepithelial lymphocytes (IEL) (only for the gut). The tissues were then cut into small pieces and incubated with Liberase (Roche) at 37°C for 30 min. After enzyme digestion, the tissue chunks were mashed through a syringe end and filtered through a 100 µm cell strainer. After washes, the cells were re-suspended in 40% Percoll, and underlaid with 80% Percoll. The cells in the interface were collected after spinning down for 20min, 800 x g at room temperature (25°C), with centrifuge brake on OFF position. The cells were cryopreserved in liquid nitrogen freezers after three washes with R10 medium. All cell samples used in this study were cryopreserved.

4. Antibodies and flow cytometric analysis of MDSC and immune activation of the tissue compartments

For flow cytometric analysis, the single-cell suspensions from different compartments were first incubated with Fc Receptor blocking reagent (Miltenyi Biotec), and then stained with viability dye (Invitrogen) and antibody mixtures including anti-CD45 (BD Pharmingen) to exclude the dead and CD45 negative cells. For immune activation, the following antibodies were used: CD3-PE-Cy7, CD4-BV605, CD8-APC-Cy7, CD14-V450, Ki67-APC, HLA-DR PE-Cy5, and CCR5-PE (BD Pharmingen); CD69-Alexa Fluor 700 (Biolegend); and CD38-FITC (STEMCELL Technologies). For MDSC analysis, the following antibodies were used: CD3-PE-Cy7, CD4-BV605, CD14-V450, Ki67-APC, HLA-DR-APC-Cy7, Lin-FITC (Lin1, BD Pharmingen); CD33-PE (Miltenyi Biotec), CD15-Alexa700 and CD11b-PE-Cy5 (Biolegend). CM9 Dextramer was obtained from Immudex (Denmark). For gag-P27 staining of MDSCs, besides the antibodies listed above, CD2-PE-Cy7 (BD Pharmingen), CD68-Perp-cy5.5, and P27 antibody [60] (clone KK64, cat#2321, from NIH AIDS reagent program) were also added to the panel. For gag-P27 staining, we have incubated the cells for 2 hrs with 5 µl of human TruX blocker (Biolegend), and 2 µl of human IgG (1 mg/ml, from R&D systems) before adding the surface antibody mixture. We also extended the P27 staining overnight on ice. An LSRII flow cytometer was used for data acquisition, and FlowJo software (Tree Star Inc.) was used for data analyses. The gating strategies for MDSCs were shown in Fig 1A and Fig 2A.

5. Flow cytometric sorting and SIV viral RNA detection

MDSCs were sorted by using a BD FACSAria (BD Biosciences) with the following markers: CD3⁻HLA-DR⁻CD11b⁺CD33⁺ cells from live PBMCs (purity greater than 95%) or spleen. CD4⁺ T cells were sorted as positive controls. The sorted CD4⁺ T cells were either serially diluted in 10-fold dilutions or directly put into Trizon. After Trizon-lysis, the sorted cells were then subjected to RNA isolation, reverse transcription, and qPCR reactions to detect the relative expression levels of SIVgag, spliced SIV rev, tat, nef, and vif, CD3, CD4, and GAPDH in MDSCs and CD4⁺ T cells (ABI, Bioline) [58]. Taqman probe and primer sets for gag[61], CD3, CD4, and GAPDH were used (ABI). Spliced SIV rev, tat, nef, and vif primers and probes were synthesized at ABI, and multiplexed to detect the rev, tat, nef, and vif targets simultaneously [62]. Each PCR mixture contained 5 µl of cDNA, 10 µl of 2X universal PCR master mix (ABI) or SensiFast probe kit (Bioline USA Inc.), 1 µl of primer/probe set. All PCR were run using ABI 7500 with program consisting of 2 min at 50°C, 10 min at 95°C, and then followed by 45 cycles of 15 sec at 95° C, and 1 min at 60°C. Relative mRNA expression levels were compared by the comparative threshold cycle (Ct) method of relative quantitation (PerkinElmer User Bulletin no. 2) as described[63], except in cases where standards were available to determine copy number, as indicated. As each cycle represents a doubling (log₂), subtracting Ct for GAPDH is essentially taking a ratio to or normalizing to the GAPDH housekeeping gene.

6. Chemotaxis assays

The 96-well ChemoTx chemotaxis system (NeuroProbe Inc., 5um pore) was used for chemotaxis assays. The lower wells were blocked with 301 µl of 1% BSA for 30 min at room temperature, which was aspirated and replaced with 301 µl of plasma from pre- or post-infected macaques as test samples. 301 µl of RPMI-1640 /0.1% BSA buffer was included in the experiments as buffer controls. Single cell suspensions from bone marrow of naive macaques were collected and 2 x 10⁵ bone marrow single cell suspensions were re-suspended in 50 µl RPMI-1640 /0.1% BSA and loaded above the membrane. After incubation for 2 hr at 37°C, 5% CO₂, the top wells were removed with a scraper and the migrated cells in the bottom wells were counted, and stained with the antibody mixture: CD3-PE-Cy7, CD14-V450, HLA-DR-APC-Cy7, Lin-FITC, CD33-PE, and CD11b-PE-Cy5, Cd45-Alexa 700, and yellow viability dye. 50 µl of CountBright absolute counting beads (Molecular Probes) were added to each tube before data acquisition using an LSRII flow cytometer, and FlowJo software (Tree Star Inc.) was used for data analyses. Chemotaxis index was defined as the ratio of the absolute cell numbers in the test samples over the absolute cell numbers in the buffer control samples.

7. Statistical analyses

We performed statistical analyses with Prism version 6 (Graph Pad). One-way ANOVA with Dunn's multiple comparison corrections, and Mann-Whitney, and Wilcoxon tests were used as shown in the figures. Spearman analysis was used for correlations. A two-sided significance level of 0.05 was used for all analyses.

Supporting information

S1 Fig. Gating strategy of CD14^{high} and CD14 ^{intermediate} MDSCs in SIV chronically infected bone marrow (14-month post-infection) and naïve bone marrow. (TIF)

S2 Fig. CD14⁺ subpopulation in SIV-infected and naive bone marrows. CD14⁺ subpopulation within the CD45⁺CD3⁻ live mononuclear cell gate from two cohorts of naïve macaques (with 28 and 12 animals separately), and one cohort of SIV chronically infected macaques (12 animals) were shown.

(TIF)

S3 Fig. The frequencies of CD14⁺ and Lin⁻CD15⁺MDSC subsets in freshly-isolated vs. cryopreserved/thawed bone marrow (BM) and PBMC samples were compared. BM and PBMC samples from SIVmac251-infected and naïve macaques were divided into two aliquots; one was immediately stained and analyzed by flow-cytometry, while the other were cryopreserved in liquid N₂ and thawed for testing 4–5 months later. SIV-infected BM (n = 6), PBMC(n = 7), and naïve BM (n = 12), PBMC (n = 5) samples were used for comparisons. (A-B) The frequencies of CD14⁺MDSCs in the BM samples did not significantly change in the SIV-infected and naïve animals, where the frequencies of Lin⁻CD15⁺MDSCs significantly decreased after cryopreservation and thawing. (C) 70–80% of CD14⁺MDSCs were maintained in the frozen naïve and SIV-infected BM samples, whereas only 20% of Lin⁻CD15⁺MDSCs were detected after cryopreservation and thawing in both the infected and naïve animals. No difference in preservation was observed between SIV-infected and naïve animals for either subset. (D-F) The frequencies of CD14⁺MDSCs in the PBMC samples did not significantly change in the SIV-infected animals after cryopreservation and thawing. The frequencies of Lin⁻CD15⁺ and CD14⁺MDSCs in the PBMC of the naïve animals, and the frequencies of Lin⁻CD15⁺ MDSCs in the PBMC of SIV-infected animals were too low to adequately assess the effect on cryopreservation, although the direction of the change in CD15⁺ Lin⁻ MDSCs in PMBC (Panel E) was in the direction of greater preservation in the SIV-infected than in the naïve populations, not consistent with any greater loss in the SIV-infected cells.

Each data point represents one individual animal. The Wilcoxon matched-pairs signed rank tests were used for comparisons.

(TIFF)

S4 Fig. The frequencies of CD14^{intermediate} and CD14^{high} MDSCs in the bone marrow of chronically SIV-infected macaques inversely correlated with plasma viral loads; but only CD14^{intermediate}, but not CD14^{high}, MDSCs positively correlated with CD4⁺ T cell preservation in the PBMC. Spearman analysis was used for correlations. (TIF)

S5 Fig. qPCR analysis of SIV-gag in the flow cytometer sorted MDSCs and CD4⁺T cells from the PBMCs of SIV-infected macaques. The Ct values of Gag, CD4, CD3, and GAPDH from the sorted MDSCs and CD4⁺T cells were shown in A&B. Equal amounts of cDNA were used to detect the expression levels of gag, CD4, CD3, and GAPDH except the CD3 in the sorted CD4⁺T cells (100-fold less cDNA was added) using macaque-specific Taqman primer/ probes. The ratio of SIVgag /CD4 in CD4⁺T cells and MDSCs were shown in C. (TIFF)

S6 Fig. qPCR analysis of SIV-gag in the flow cytometer sorted MDSCs and CD4⁺T cells from the PBMCs of SIV-infected macaques. The Ct values of Gag, CD4, and GAPDH from the sorted MDSCs and CD4⁺T cells were shown in A-C. Equal amounts of cDNA were used to detect the expression levels of gag, and CD4 using macaque-specific Taqman primer/probes. The ratio of SIVgag expression level in MDSCs vs SIV gag expression level in CD4⁺T cells is shown in D, calculating using 2^ (Δgag-GAPDH) for MDSCs vs CD4⁺T cells. The ratio of CD4 expression level in MDSCs vs CD4 expression level in CD4⁺T cells were shown also in D using the similar calculation method. MDSCs had a 50-fold lower expression level of CD4. Thus, the level of CD4⁺T cell contamination in the MDSC population is 50-fold too low to account for the SIVgag present. (TIF)

S7 Fig. qPCR analysis of SIV mRNA in the flow cytometer sorted MDSCs and CD4⁺ T cells from the PBMC or BM samples of the SIV-infected macaques. Three independent experiments are shown. MDSCs and CD4⁺ T cells from the PBMC or bone marrow samples of the SIVmac251 chronically infected animals were sorted as described in Fig 5A. After RNA isolation and cDNA synthesis, each of the CD4⁺ T and MDSC samples was subjected to qPCR analysis for SIVgag and/or spliced SIV mRNA, CD3, or CD4, and GAPDH using the same amounts of cDNA for that sample. (A-B) In this experiment, four PBMC specimens from the SIVmac251-infected macaques were analyzed for unspliced SIVgag and spliced SIV mRNA. The threshold cycles of spliced and unspliced viral mRNA in T cells and MDSCs of the infected macaques are shown in A-B. Lower threshold cycle value indicates higher expression level of the target sequence. For CD3, if there was no amplification signal after 45 cycles, Ct value of 45 was assigned to the tested sample. (C) In another independent experiment, MDSCs and $CD4^+ T$ cells from the bone marrow samples of the six SIV-infected animals were sorted. For the sorted $CD4^+ T$ cells, the samples were serially diluted (10–fold) before the RNA isolation. Spliced SIV mRNA and CD3 mRNA from six MDSC samples and one example of $CD4^+ T$ cells were analyzed. The detection limit is 90 to 900000 $CD4^+ T$ cells. Of the 4 animals with no detectable CD3 contamination (shown in magenta squares), 3 had clear spliced mRNA expression. (D) In the third independent experiment, MDSCs and CD4⁺ T cells from the PBMC of six SIV-infected animals were sorted. For the sorted CD4⁺ T cells, the samples were serially diluted (10–fold) before the RNA isolation. Gag and CD4 were shown in Fig 5D. For CD3, if there was no amplification signal after 45 cycles, Ct value of 45 was assigned to the tested sample. Black dots in C are CD4⁺ T cells as positive control titration curve, while red or magenta squares are the MDSCs plotted on the same scale. Black dots in D are CD4⁺ T cells as positive control titration curve, while red squares are the MDSCs plotted on the same scale. Black dots in D are CD4⁺ T cells as positive to detect T cell contamination in these four animals was from 27 and 118 CD4⁺ T cells.

(TIFF)

S8 Fig. Flow cytometric detection of gag-P27⁺ MDSCs and CD4⁺T cells in the bone marrow of SIV-infected or naive animals. Flow cytometric plots of P27 staining (A), and summary of P27⁺ % within CD4⁺T cells, and MDSCs (B) from one experiment with 1 SIV-infected and 4 naïve animals were shown.



S9 Fig. Decreased Lin⁻ CD15⁺MDSCs in bone marrow were not associated with increased immune activation except in PBMC. Spearman analysis was used for correlations. The correlations between the frequency of Lin⁻ CD15⁺MDSCs and the Ki67⁺CD8⁺ T cell frequencies in the SIV-infected bone marrow (A), PBMC (B), spleen (D), Axillary LN (E) and Mesenteric LN (F), but not liver (C), ileum LP (G) or colon LP (H) were shown. Each data point represents one animal.

(TIF)

S10 Fig. The frequencies of MDSCs were initially significantly increased 4 months post-SHIV_{SF162P4} infection, and reverted to normal 11 months post-infection. (TIF)

S1 Table. Information on the macaques used in this study. (PPTX)

S2 Table. The R and P values of Spearman's analysis among the CD14+MDSC from different compartments. (PPTX)

S3 Table. The R and P values of Spearman's analysis among the Lin-CD15+MDSC from different compartments.

(PPTX)

S4 Table. The R and P values of Spearman's analysis among the CM9 Dextramer+ frequency within the total CD8+T cells from different compartments. (PPTX)

S5 Table. The R and P values of Spearman's analysis among the Ki67⁺ frequency within the total CD8+T cells from different compartments. (PPTX)

Acknowledgments

The following reagent was obtained through the NIH AIDS Reagent Program, Division of AIDS, NIAID, NIH: Anti-SIVmac251 p27 Monoclonal (KK64) from Dr. Karen Kent and Miss Caroline Powell. We would like to thank Joe Dutra from University of California Davis for providing SIVgag standard for qPCR; Dr. Tim Greten from National Cancer Center, NIH for advice on MDSC gating.

Author Contributions

Conceptualization: YS JAB CJM.

Data curation: YS BF YW RB SK TR LF KM.

Formal analysis: YS.

Writing - original draft: YS.

Writing - review & editing: YS JAB CJM.

References

- Almand B, Clark JI, Nikitina E, van Beynen J, English NR, Knight SC, et al. Increased production of immature myeloid cells in cancer patients: a mechanism of immunosuppression in cancer. Journal of immunology. 2001; 166(1):678–89.
- Kusmartsev S, Gabrilovich DI. Role of immature myeloid cells in mechanisms of immune evasion in cancer. Cancer immunology, immunotherapy: CII. 2006; 55(3):237–45. PubMed Central PMCID: PMC1350971. https://doi.org/10.1007/s00262-005-0048-z PMID: 16047143
- Filipazzi P, Huber V, Rivoltini L. Phenotype, function and clinical implications of myeloid-derived suppressor cells in cancer patients. Cancer immunology, immunotherapy: CII. 2012; 61(2):255–63. https://doi.org/10.1007/s00262-011-1161-9 PMID: 22120756
- Zhao F, Obermann S, von Wasielewski R, Haile L, Manns MP, Korangy F, et al. Increase in frequency of myeloid-derived suppressor cells in mice with spontaneous pancreatic carcinoma. Immunology. 2009; 128(1):141–9. PubMed Central PMCID: PMC2747147. <u>https://doi.org/10.1111/j.1365-2567.</u> 2009.03105.x PMID: 19689743
- Ilkovitch D, Lopez DM. The liver is a site for tumor-induced myeloid-derived suppressor cell accumulation and immunosuppression. Cancer research. 2009; 69(13):5514–21. PubMed Central PMCID: PMC2706931. https://doi.org/10.1158/0008-5472.CAN-08-4625 PMID: 19549903
- Kusmartsev S, Gabrilovich DI. Inhibition of myeloid cell differentiation in cancer: the role of reactive oxygen species. Journal of leukocyte biology. 2003; 74(2):186–96. PMID: 12885935
- Porembka MR, Mitchem JB, Belt BA, Hsieh CS, Lee HM, Herndon J, et al. Pancreatic adenocarcinoma induces bone marrow mobilization of myeloid-derived suppressor cells which promote primary tumor growth. Cancer immunology, immunotherapy: CII. 2012; 61(9):1373–85. PubMed Central PMCID: PMC3697836. https://doi.org/10.1007/s00262-011-1178-0 PMID: 22215137
- Gabrilovich DI, Nagaraj S. Myeloid-derived suppressor cells as regulators of the immune system. Nature reviews Immunology. 2009; 9(3):162–74. PubMed Central PMCID: PMC2828349. <u>https://doi.org/10.1038/nri2506 PMID: 19197294</u>
- Elkabets M, Ribeiro VS, Dinarello CA, Ostrand-Rosenberg S, Di Santo JP, Apte RN, et al. IL-1beta regulates a novel myeloid-derived suppressor cell subset that impairs NK cell development and function. European journal of immunology. 2010; 40(12):3347–57. PubMed Central PMCID: PMCPMC3373225. https://doi.org/10.1002/eji.201041037 PMID: 21110318
- Gabrilovich DI, Ostrand-Rosenberg S, Bronte V. Coordinated regulation of myeloid cells by tumours. Nature reviews Immunology. 2012; 12(4):253–68. PubMed Central PMCID: PMCPMC3587148. <u>https://doi.org/10.1038/nri3175 PMID: 22437938</u>
- Parker KH, Beury DW, Ostrand-Rosenberg S. Myeloid-Derived Suppressor Cells: Critical Cells Driving Immune Suppression in the Tumor Microenvironment. Adv Cancer Res. 2015; 128:95–139. PubMed Central PMCID: PMCPMC4662416. https://doi.org/10.1016/bs.acr.2015.04.002 PMID: 26216631

- Vollbrecht T, Stirner R, Tufman A, Roider J, Huber RM, Bogner JR, et al. Chronic progressive HIV-1 infection is associated with elevated levels of myeloid-derived suppressor cells. Aids. 2012; 26(12): F31–7. https://doi.org/10.1097/QAD.0b013e328354b43f PMID: 22526518
- Garg A, Spector SA. HIV type 1 gp120-induced expansion of myeloid derived suppressor cells is dependent on interleukin 6 and suppresses immunity. The Journal of infectious diseases. 2014; 209(3):441–51. PubMed Central PMCID: PMC3883171. https://doi.org/10.1093/infdis/jit469 PMID: 23999600
- 14. Wang L, Zhao J, Ren JP, Wu XY, Morrison ZD, Elgazzar MA, et al. Expansion of myeloid-derived suppressor cells promotes differentiation of regulatory T cells in HIV-1+ individuals. Aids. 2016.
- Ostrand-Rosenberg S. Myeloid-derived suppressor cells: more mechanisms for inhibiting antitumor immunity. Cancer immunology, immunotherapy: CII. 2010; 59(10):1593–600. PubMed Central PMCID: PMCPMC3706261. https://doi.org/10.1007/s00262-010-0855-8 PMID: 20414655
- Hoechst B, Ormandy LA, Ballmaier M, Lehner F, Kruger C, Manns MP, et al. A new population of myeloid-derived suppressor cells in hepatocellular carcinoma patients induces CD4(+)CD25(+)Foxp3(+) T cells. Gastroenterology. 2008; 135(1):234–43. <u>https://doi.org/10.1053/j.gastro.2008.03.020</u> PMID: 18485901
- Solito S, Marigo I, Pinton L, Damuzzo V, Mandruzzato S, Bronte V. Myeloid-derived suppressor cell heterogeneity in human cancers. Ann N Y Acad Sci. 2014; 1319:47–65. https://doi.org/10.1111/nyas. 12469 PMID: 24965257
- Bronte V, Brandau S, Chen SH, Colombo MP, Frey AB, Greten TF, et al. Recommendations for myeloid-derived suppressor cell nomenclature and characterization standards. Nat Commun. 2016; 7:12150. PubMed Central PMCID: PMCPMC4935811. https://doi.org/10.1038/ncomms12150 PMID: 27381735
- Greten TF, Manns MP, Korangy F. Myeloid derived suppressor cells in human diseases. International immunopharmacology. 2011; 11(7):802–7. PubMed Central PMCID: PMC3478130. <u>https://doi.org/10.1016/j.intimp.2011.01.003</u> PMID: 21237299
- Youn JI, Kumar V, Collazo M, Nefedova Y, Condamine T, Cheng P, et al. Epigenetic silencing of retinoblastoma gene regulates pathologic differentiation of myeloid cells in cancer. Nat Immunol. 2013; 14 (3):211–20. PubMed Central PMCID: PMCPMC3578019. <u>https://doi.org/10.1038/ni.2526</u> PMID: 23354483
- Haverkamp JM, Smith AM, Weinlich R, Dillon CP, Qualls JE, Neale G, et al. Myeloid-derived suppressor activity is mediated by monocytic lineages maintained by continuous inhibition of extrinsic and intrinsic death pathways. Immunity. 2014; 41(6):947–59. Central PMCID: PMCPMC4272664. https://doi.org/10. 1016/j.immuni.2014.10.020 PMID: 25500368
- Qin A, Cai W, Pan T, Wu K, Yang Q, Wang N, et al. Expansion of monocytic myeloid-derived suppressor cells dampens T cell function in HIV-1-seropositive individuals. Journal of virology. 2013; 87 (3):1477–90. PubMed Central PMCID: PMC3554138. <u>https://doi.org/10.1128/JVI.01759-12</u> PMID: 23152536
- Sui Y, Hogg A, Wang Y, Frey B, Yu H, Xia Z, et al. Vaccine-induced myeloid cell population dampens protective immunity to SIV. The Journal of clinical investigation. 2014; 124(6):2538–49. PubMed Central PMCID: PMC4038576. https://doi.org/10.1172/JCI73518 PMID: 24837435
- Tacke RS, Tosello-Trampont A, Nguyen V, Mullins DW, Hahn YS. Extracellular hepatitis C virus core protein activates STAT3 in human monocytes/macrophages/dendritic cells via an IL-6 autocrine pathway. The Journal of biological chemistry. 2011; 286(12):10847–55. PubMed Central PMCID: PMC3060535. https://doi.org/10.1074/jbc.M110.217653 PMID: 21282107
- Cai W, Qin A, Guo P, Yan D, Hu F, Yang Q, et al. Clinical significance and functional studies of myeloidderived suppressor cells in chronic hepatitis C patients. Journal of clinical immunology. 2013; 33 (4):798–808. https://doi.org/10.1007/s10875-012-9861-2 PMID: 23354838
- Fortin C, Huang X, Yang Y. NK cell response to vaccinia virus is regulated by myeloid-derived suppressor cells. Journal of immunology. 2012; 189(4):1843–9. PubMed Central PMCID: PMC3411873.
- Zhu J, Huang X, Yang Y. Myeloid-derived suppressor cells regulate natural killer cell response to adenovirus-mediated gene transfer. Journal of virology. 2012; 86(24):13689–96. PubMed Central PMCID: PMC3503091. https://doi.org/10.1128/JVI.01595-12 PMID: 23055553
- Liu C, Zhang C, Lu H, Cai J, Wang Z, Chen J, et al. Poly(I:C) induce bone marrow precursor cells into myeloid-derived suppressor cells. Molecular and cellular biochemistry. 2011; 358(1–2):317–23. <u>https://</u> doi.org/10.1007/s11010-011-0982-3 PMID: 21744070
- Gallina G, Dolcetti L, Serafini P, De Santo C, Marigo I, Colombo MP, et al. Tumors induce a subset of inflammatory monocytes with immunosuppressive activity on CD8+ T cells. The Journal of clinical investigation. 2006; 116(10):2777–90. PubMed Central PMCID: PMC1578632. <u>https://doi.org/10.1172/</u> JCI28828 PMID: 17016559

- Delano MJ, Scumpia PO, Weinstein JS, Coco D, Nagaraj S, Kelly-Scumpia KM, et al. MyD88-dependent expansion of an immature GR-1(+)CD11b(+) population induces T cell suppression and Th2 polarization in sepsis. The Journal of experimental medicine. 2007; 204(6):1463–74. PubMed Central PMCID: PMC2118626. https://doi.org/10.1084/jem.20062602 PMID: 17548519
- Brudecki L, Ferguson DA, McCall CE, El Gazzar M. Myeloid-derived suppressor cells evolve during sepsis and can enhance or attenuate the systemic inflammatory response. Infection and immunity. 2012; 80(6):2026–34. PubMed Central PMCID: PMC3370575. <u>https://doi.org/10.1128/IAI.00239-12</u> PMID: 22451518
- Diaz-Montero CM, Salem ML, Nishimura MI, Garrett-Mayer E, Cole DJ, Montero AJ. Increased circulating myeloid-derived suppressor cells correlate with clinical cancer stage, metastatic tumor burden, and doxorubicin-cyclophosphamide chemotherapy. Cancer immunology, immunotherapy: CII. 2009; 58 (1):49–59. PubMed Central PMCID: PMC3401888. <u>https://doi.org/10.1007/s00262-008-0523-4</u> PMID: 18446337
- 33. Mundy-Bosse BL, Young GS, Bauer T, Binkley E, Bloomston M, Bill MA, et al. Distinct myeloid suppressor cell subsets correlate with plasma IL-6 and IL-10 and reduced interferon-alpha signaling in CD4(+) T cells from patients with GI malignancy. Cancer immunology, immunotherapy: CII. 2011; 60(9):1269–79. PubMed Central PMCID: PMC3521517. https://doi.org/10.1007/s00262-011-1029-z PMID: 21604071
- 34. Gabitass RF, Annels NE, Stocken DD, Pandha HA, Middleton GW. Elevated myeloid-derived suppressor cells in pancreatic, esophageal and gastric cancer are an independent prognostic factor and are associated with significant elevation of the Th2 cytokine interleukin-13. Cancer immunology, immuno-therapy: CII. 2011; 60(10):1419–30. PubMed Central PMCID: PMC3176406. https://doi.org/10.1007/s00262-011-1028-0 PMID: 21644036
- 35. Allen TM, Mothe BR, Sidney J, Jing P, Dzuris JL, Liebl ME, et al. CD8(+) lymphocytes from simian immunodeficiency virus-infected rhesus macaques recognize 14 different epitopes bound by the major histocompatibility complex class I molecule manu-A*01: implications for vaccine design and testing. Journal of virology. 2001; 75(2):738–49. PubMed Central PMCID: PMCPMC113970. https://doi.org/10. 1128/JVI.75.2.738-749.2001 PMID: 11134287
- Ezernitchi AV, Vaknin I, Cohen-Daniel L, Levy O, Manaster E, Halabi A, et al. TCR zeta down-regulation under chronic inflammation is mediated by myeloid suppressor cells differentially distributed between various lymphatic organs. Journal of immunology. 2006; 177(7):4763–72.
- Kotsakis A, Harasymczuk M, Schilling B, Georgoulias V, Argiris A, Whiteside TL. Myeloid-derived suppressor cell measurements in fresh and cryopreserved blood samples. J Immunol Methods. 2012; 381 (1–2):14–22. PubMed Central PMCID: PMCPMC3385927. <u>https://doi.org/10.1016/j.jim.2012.04.004</u> PMID: 22522114
- Calantone N, Wu F, Klase Z, Deleage C, Perkins M, Matsuda K, et al. Tissue myeloid cells in SIVinfected primates acquire viral DNA through phagocytosis of infected T cells. Immunity. 2014; 41 (3):493–502. PubMed Central PMCID: PMCPMC4241569. https://doi.org/10.1016/j.immuni.2014.08. 014 PMID: 25238099
- 39. Fan Q, Gu D, Liu H, Yang L, Zhang X, Yoder MC, et al. Defective TGF-beta signaling in bone marrowderived cells prevents hedgehog-induced skin tumors. Cancer research. 2014; 74(2):471–83. PubMed Central PMCID: PMC3963525. https://doi.org/10.1158/0008-5472.CAN-13-2134-T PMID: 24282281
- Ansari AW, Heiken H, Meyer-Olson D, Schmidt RE. CCL2: a potential prognostic marker and target of anti-inflammatory strategy in HIV/AIDS pathogenesis. European journal of immunology. 2011; 41 (12):3412–8. https://doi.org/10.1002/eji.201141676 PMID: 22076814
- **41.** Isgro A, Aiuti A, Leti W, Gramiccioni C, Esposito A, Mezzaroma I, et al. Immunodysregulation of HIV disease at bone marrow level. Autoimmun Rev. 2005; 4(8):486–90. https://doi.org/10.1016/j.autrev.2005. 04.014 PMID: 16214083
- 42. Alexaki A, Wigdahl B. HIV-1 infection of bone marrow hematopoietic progenitor cells and their role in trafficking and viral dissemination. PLoS pathogens. 2008; 4(12):e1000215. PubMed Central PMCID: PMCPMC2603331. https://doi.org/10.1371/journal.ppat.1000215 PMID: 19112504
- Vishnu P, Aboulafia DM. Haematological manifestations of human immune deficiency virus infection. Br J Haematol. 2015; 171(5):695–709. https://doi.org/10.1111/bjh.13783 PMID: 26452169
- Bahner I, Kearns K, Coutinho S, Leonard EH, Kohn DB. Infection of human marrow stroma by human immunodeficiency virus-1 (HIV-1) is both required and sufficient for HIV-1-induced hematopoietic suppression in vitro: demonstration by gene modification of primary human stroma. Blood. 1997; 90 (5):1787–98. PMID: 9292511
- Kulkosky J, Bouhamdan M, Geist A, Nunnari G, Phinney DG, Pomerantz RJ. Pathogenesis of HIV-1 infection within bone marrow cells. Leukemia & lymphoma. 2000; 37(5–6):497–515.

- Goh C, Narayanan S, Hahn YS. Myeloid-derived suppressor cells: the dark knight or the joker in viral infections? Immunological reviews. 2013; 255(1):210–21. PubMed Central PMCID: PMC3748397. https://doi.org/10.1111/imr.12084 PMID: 23947357
- 47. Chahroudi A, Cartwright E, Lee ST, Mavigner M, Carnathan DG, Lawson B, et al. Target cell availability, rather than breast milk factors, dictates mother-to-infant transmission of SIV in sooty mangabeys and rhesus macaques. PLoS pathogens. 2014; 10(3):e1003958. PubMed Central PMCID: PMC3946396. https://doi.org/10.1371/journal.ppat.1003958 PMID: 24604066
- Mehraj V, Jenabian MA, Vyboh K, Routy JP. Immune Suppression by Myeloid Cells in HIV Infection: New Targets for Immunotherapy. The open AIDS journal. 2014; 8:66–78. PubMed Central PMCID: PMC4302459. https://doi.org/10.2174/1874613601408010066 PMID: 25624956
- Brenchley JM, Price DA, Douek DC. HIV disease: fallout from a mucosal catastrophe? Nat Immunol. 2006; 7(3):235–9. https://doi.org/10.1038/ni1316 PMID: 16482171
- Brenchley JM, Price DA, Schacker TW, Asher TE, Silvestri G, Rao S, et al. Microbial translocation is a cause of systemic immune activation in chronic HIV infection. Nat Med. 2006; 12(12):1365–71. https://doi.org/10.1038/nm1511 PMID: 17115046
- Youn JI, Gabrilovich DI. The biology of myeloid-derived suppressor cells: the blessing and the curse of morphological and functional heterogeneity. European journal of immunology. 2010; 40(11):2969–75. PubMed Central PMCID: PMC3277452. https://doi.org/10.1002/eji.201040895 PMID: 21061430
- 52. Dai J, El Gazzar M, Li GY, Moorman JP, Yao ZQ. Myeloid-derived suppressor cells: paradoxical roles in infection and immunity. Journal of innate immunity. 2015; 7(2):116–26. PubMed Central PMCID: PMC4348209. https://doi.org/10.1159/000368233 PMID: 25401944
- Seddiki N, Brezar V, Draenert R. Cell exhaustion in HIV-1 infection: role of suppressor cells. Current opinion in HIV and AIDS. 2014; 9(5):452–8. https://doi.org/10.1097/COH.0000000000087 PMID: 25010895
- Sinha P, Clements VK, Bunt SK, Albelda SM, Ostrand-Rosenberg S. Cross-talk between myeloidderived suppressor cells and macrophages subverts tumor immunity toward a type 2 response. Journal of immunology. 2007; 179(2):977–83.
- Azzaoui I, Uhel F, Rossille D, Pangault C, Dulong J, Le Priol J, et al. T-cell defect in diffuse large B-cell lymphomas involves expansion of myeloid-derived suppressor cells. Blood. 2016; 128(8):1081–92. https://doi.org/10.1182/blood-2015-08-662783 PMID: 27338100
- Moses AV, Williams S, Heneveld ML, Strussenberg J, Rarick M, Loveless M, et al. Human immunodeficiency virus infection of bone marrow endothelium reduces induction of stromal hematopoietic growth factors. Blood. 1996; 87(3):919–25. PMID: 8562963
- Sui Y, Lee EM, Wang Y, Hogg A, Frey B, Venzon D, et al. Early SIV dissemination after intra-rectal SIVmac251 challenge was associated with proliferating virus-susceptible cells in the colorectum. Journal of acquired immune deficiency syndromes. 2015.
- Sui Y, Zhu Q, Gagnon S, Dzutsev A, Terabe M, Vaccari M, et al. Innate and adaptive immune correlates of vaccine and adjuvant-induced control of mucosal transmission of SIV in macaques. Proceedings of the National Academy of Sciences of the United States of America. 2010; 107(21):9843–8. PubMed Central PMCID: PMC2906837. https://doi.org/10.1073/pnas.0911932107 PMID: 20457926
- Sui Y, Gagnon S, Dzutsev A, Zhu Q, Yu H, Hogg A, et al. TLR agonists and/or IL-15 adjuvanted mucosal SIV vaccine reduced gut CD4(+) memory T cell loss in SIVmac251-challenged rhesus macaques. Vaccine. 2011; 30(1):59–68. PubMed Central PMCID: PMC3258186. https://doi.org/10.1016/j.vaccine. 2011.10.043 PMID: 22041305
- Kent KA, Gritz L, Stallard G, Cranage MP, Collignon C, Thiriart C, et al. Production and of monoclonal antibodies to simian immunodeficiency virus envelope glycoproteins. Aids. 1991; 5(7):829–36. PMID: 1716442
- Leutenegger CM, Higgins J, Matthews TB, Tarantal AF, Luciw PA, Pedersen NC, et al. Real-time Taq-Man PCR as a specific and more sensitive alternative to the branched-chain DNA assay for quantitation of simian immunodeficiency virus RNA. AIDS Res Hum Retroviruses. 2001; 17(3):243–51. https://doi. org/10.1089/088922201750063160 PMID: 11177407
- Ma ZM, Dutra J, Fritts L, Miller CJ. Lymphatic Dissemination of Simian Immunodeficiency Virus after Penile Inoculation. Journal of virology. 2016; 90(8):4093–104. PubMed Central PMCID: PMCPMC4810538. https://doi.org/10.1128/JVI.02947-15 PMID: 26865706
- 63. Godfrey TE, Kim SH, Chavira M, Ruff DW, Warren RS, Gray JW, et al. Quantitative mRNA expression analysis from formalin-fixed, paraffin-embedded tissues using 5' nuclease quantitative reverse transcription-polymerase chain reaction. J Mol Diagn. 2000; 2(2):84–91. PubMed Central PMCID: PMCPMC1906896. https://doi.org/10.1016/S1525-1578(10)60621-6 PMID: 11272893